Author's response to reviews

Title: Cystic renal cell carcinoma: a report of 67 cases including 4 cases with concurrent renal cell carcinoma

Authors:

Shanwen Chen (chensw123@126.com)
Baiye Jin (jinbaiye1964@aliyun.com)
Liqi Xu (urology123@hotmail.com)
Guohou Fu (fuguanghou@126.com)
Hongzhou Meng (menghz@126.com)
Ben Liu (drliuben@sina.com)
Jun Li (csw123@sohu.com)
Dan Xia (xiaxy1998@163.com)

Version: 4
Date: 5 September 2014

Author's response to reviews: see over
Date Sep 5, 2014

Dear Clare Partridge

Thank you for your acceptance of our manuscript MS: 1060147114129920 entitled “Cystic renal cell carcinoma: a report of 67 cases including 4 cases of concurrent renal cell carcinomas”. We are particularly grateful for the insightful comments of you, which greatly helped us in revision. Based on their and your comments, we have revised our manuscript.

Below please find their and your comments and our point-by-point responses. Thank you again for the revision of our manuscript. Should you need any additional information, please do not hesitate to contact me.

Sincerely,
Shanwen Chen, M.D
Editors' comments and authors’ responses:

1. Patient identifiers
Please be aware that there is information present on the images in figure 1 that would allow patients to be identified. Please modify these images so that now identifiers are present e.g. date/time/patient age etc.

Answer: Thank you for your reminding, we have modified these images so that now identifiers are not present e.g. date/time/patient age etc.

2. Abstract
In accordance with our Instructions for Authors we would ask that you include a brief description of the context of the study in the Background of the Abstract.

Answer: Thank you for the suggestions! In accordance with your Instructions for Authors we have included a brief description of the context of the study in the Background of the Abstract.

3. Title
Please correct the title of your manuscript, both on our submission system and within the pdf, as it presently contains an error: 'carcinom'.

Answer: Thank you for your reminding, this is our fault, and we have corrected the title of our manuscript (Cystic renal cell carcinoma: a report of 67 cases including 4 cases with concurrent renal cell carcinoma).